Author's response to reviews

Title: Study on Salvinolic acid B in reduction of epidural fibrosis in laminectomy rats

Authors:

Feng Chen (fengchen2016@126.com)
Zhenbo Zuo (zhenbozuo2015@126.com)
Kai Wang (kaiwang2015@126.com)
Chengdong Zhang (chengdongzhang2015@126.com)
Haifeng Gong (haifenggong2015@126.com)
Fagang Ye (fagangye2015@126.com)
Aiyu Ji (aiyu ji2015@126.com)
Hao Tao (haotao2015@126.com)

Version: 2 Date: 27 June 2014

Author's response to reviews:

1. I added the line number on each page in the present version.
2. The acknowledgement section is inserted as per editor suggestion.